

Direct mosquito feedings on dengue-2 virus-infected people reveal dynamics of human infectiousness

Louis Lambrechts, Robert C Reiner, M Veronica Briesemeister, Patricia Barrera, Kanya C Long, William H Elson, Alfonso Vizcarra, Helvio Astete, Isabel Bazan, Crystyan Siles, et al.

▶ To cite this version:

Louis Lambrechts, Robert C Reiner, M Veronica Briesemeister, Patricia Barrera, Kanya C Long, et al.. Direct mosquito feedings on dengue-2 virus-infected people reveal dynamics of human infectiousness. PLoS Neglected Tropical Diseases, 2023, 17 (9), pp.e0011593. 10.1371/journal.pntd.0011593. pasteur-04569949

HAL Id: pasteur-04569949 https://pasteur.hal.science/pasteur-04569949

Submitted on 6 May 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License



Citation: Lambrechts L, Reiner RC, Jr, Briesemeister MV, Barrera P, Long KC, Elson WH, et al. (2023) Direct mosquito feedings on dengue-2 virus-infected people reveal dynamics of human infectiousness. PLoS Negl Trop Dis 17(9): e0011593. https://doi.org/10.1371/journal. pntd.0011593

Editor: Elvina Viennet, Australian Red Cross Lifelood, AUSTRALIA

Received: March 26, 2023

Accepted: August 14, 2023

Published: September 1, 2023

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0011593

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: R code and data used for statistical analysis and to produce figures in the manuscript are openly available on GitHub [https:// RESEARCH ARTICLE

Direct mosquito feedings on dengue-2 virusinfected people reveal dynamics of human infectiousness

Louis Lambrechts¹[•]*, Robert C. Reiner, Jr[•]²[•]*, M. Veronica Briesemeister³, Patricia Barrera^{3,4}, Kanya C. Long⁵, William H. Elson³, Alfonso Vizcarra³, Helvio Astete^{6,7}, Isabel Bazan⁶, Crystyan Siles⁶, Stalin Vilcarromero^{6¤a}, Mariana Leguia⁴, Anna B. Kawiecki⁸, T. Alex Perkins⁹, Alun L. Lloyd¹⁰, Lance A. Waller¹¹, Uriel Kitron¹², Sarah A. Jenkins^{6¤b}, Robert D. Hontz^{6¤c}, Wesley R. Campbell^{6¤d}, Lauren B. Carrington¹³, Cameron P. Simmons¹⁴, J. Sonia Ampuero⁶, Gisella Vasquez⁷, John P. Elder¹⁵, Valerie A. Paz-Soldan¹⁶, Gonzalo M. Vazquez-Prokopec¹², Alan L. Rothman¹⁷, Christopher M. Barker⁸, Thomas W. Scott^{3‡}, Amy C. Morrison^{8‡}

1 Institut Pasteur, Université Paris Cité, CNRS UMR2000, Insect-Virus Interactions Unit, Paris, France, 2 University of Washington, Seattle, Washington, United States of America, 3 Department of Entomology and Nematology, University of California, Davis, California, United States of America, 4 Genomics Laboratory, Pontificia Universidad Católica del Peru, Lima, Peru, 5 Department of Family Medicine and Public Health, University of California San Diego School of Medicine, La Jolla, California, United States of America, 6 Virology and Emerging Infections Department, United States Naval Medical Research Unit No. 6, Lima, Peru, 7 Department of Entomology, United States Naval Medical Research Unit No. 6, Lima, Peru, 8 Department of Pathology, Microbiology, and Immunology, School of Veterinary Medicine, University of California, Davis, California, United States of America, 9 Department of Biological Sciences and Eck Institute for Global Health, University of Notre Dame, Notre Dame, Indiana, United States of America, 10 Biomathematics Graduate Program and Department of Mathematics, North Carolina State University, Raleigh, North Carolina, United States of America, 11 Department of Biostatistics and Bioinformatics, Rollins School of Public Health, Emory University, Atlanta, Georgia, United States of America, 12 Department of Environmental Sciences, Emory University, Atlanta, Georgia, United States of America, 13 Global Malaria Program, World Health Organization, Geneva, Switzerland, 14 Institute for Vector-Borne Disease, Monash University, Clayton, Victoria, Australia, 15 School of Public Health, San Diego State University, San Diego, California, United States of America, 16 Department of Tropical Medicine, Tulane University School of Public Health and Tropical Medicine, New Orleans, Louisiana, United States of America, 17 Institute for Immunology and Informatics and Department of Cell and Molecular Biology, University of Rhode Island, Providence, Rhode Island, United States of America

• These authors contributed equally to this work.

¤a Current address: Hospital Nacional Edgardo Rebagliati Martins, ESSALUD Lima, Peru

¤b Current address: Naval Medical Research Command Silver Spring, Maryland, United States of America ¤c Current address: Department of Emerging Infectious Diseases, United States Naval Medical Research Unit TWO, Singapore, Singapore

¤d Current address: Department of Medicine, Walter Reed National Military Medical Center, Bethesda, Maryland, United States of America

‡ TWS and ACM also contributed equally to this work.

* louis.lambrechts@pasteur.fr (LL); bcreiner@uw.edu (RCR)

Abstract

Dengue virus (DENV) transmission from humans to mosquitoes is a poorly documented, but critical component of DENV epidemiology. Magnitude of viremia is the primary determinant of successful human-to-mosquito DENV transmission. People with the same level of viremia, however, can vary in their infectiousness to mosquitoes as a function of other factors that remain to be elucidated. Here, we report on a field-based study in the city of Iquitos, github.com/bcreiner/mosq_feeds_lquitos] and provided as a supplementary file (S1 File), respectively.

Funding: The study was primarily funded by a grant from the U.S. National Institutes of Health/ National Institute of Allergy and Infectious Diseases (P01 AI098670 to TWS) with some additional salary support provided by U01AI151814 to ACM. The study was also supported by the European Union's Horizon 2020 research and innovation programme under ZikaPLAN (734584 to LL) and the French Government's Investissement d'Avenir program Laboratoire d'Excellence Integrative Biology of Emerging Infectious Diseases (ANR-10-LABX-62-IBEID to LL). Further support was provided by a sub contract from the Bill and Melinda Gates Foundation to the University of Notre Dame (OPP1081737 to TWS/ACM), the Defense Threat Reduction Agency, Military Infectious Disease Research Program (MIDRP, S0520 15 LI and S0572 17 LI), and the Armed Forces Health Surveillance Division. Global Emerging Infections System Branch (ProMIS ID: 20160390169, P0090_17_N6_1.1.1, P0106_18_N6_01.01, and P0143_19_N6). The views expressed in this article are those of the authors and do not necessarily reflect the official policy or position of the Department of the Navy. Department of Defense, nor the U.S. Government. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Peru, where we conducted direct mosquito feedings on people naturally infected with DENV and that experienced mild illness. We also enrolled people naturally infected with Zika virus (ZIKV) after the introduction of ZIKV in Iquitos during the study period. Of the 54 study participants involved in direct mosquito feedings, 43 were infected with DENV-2, two with DENV-3, and nine with ZIKV. Our analysis excluded participants whose viremia was detectable at enrollment but undetectable at the time of mosquito feeding, which was the case for all participants with DENV-3 and ZIKV infections. We analyzed the probability of onward transmission during 50 feeding events involving 27 participants infected with DENV-2 based on the presence of infectious virus in mosquito saliva 7-16 days post blood meal. Transmission probability was positively associated with the level of viremia and duration of extrinsic incubation in the mosquito. In addition, transmission probability was influenced by the day of illness in a non-monotonic fashion; i.e., transmission probability increased until 2 days after symptom onset and decreased thereafter. We conclude that mildly ill DENV-infected humans with similar levels of viremia during the first two days after symptom onset will be most infectious to mosquitoes on the second day of their illness. Quantifying variation within and between people in their contribution to DENV transmission is essential to better understand the biological determinants of human infectiousness, parametrize epidemiological models, and improve disease surveillance and prevention strategies.

Author summary

In this study, we examined the potential for people with mild illness to transmit dengue virus to mosquitoes. Although people with mild dengue illness can be infectious, they often are undetected by surveillance systems and research teams. Variation in infectiousness over the course of a person's infection is largely unexplored, but potentially important component of dengue epidemiology. In an effort to better understand the factors that underlie this variation, we blood fed laboratory-reared, uninfected mosquitoes directly on mildly ill people naturally infected with dengue virus in Iquitos, Peru. Detection of virus in mosquito saliva after feeding on an infected person was considered positive for mosquito transmission. As has been reported previously, transmission to mosquitoes was positively associated with the amount of virus in a person's blood and the number of days after the mosquito fed on an infected person. Unlike results from previous studies, which reported peak transmission during the first day of a person's illness, we found that for mildly ill people infectiousness to mosquitoes peaked on the second day of illness. Because people with mild illness are thought to contribute more to dengue virus transmission than more severely ill people, subtle differences in the dynamics of their infectiousness to mosquitoes could be epidemiologically significant.

Introduction

Dengue is one of the fastest-growing global infectious diseases [1]. It is now endemic in an increasing number of tropical cities and its geographical range is expected to further expand due to ongoing global phenomena including urbanization and climate change [2-4]. There are an estimated 400 million yearly human dengue virus (DENV) infections, of which approximately 100 million are apparent with symptoms [5]. Dengue is a self-limited infection with a

broad range of clinical manifestations ranging from asymptomatic infection to life-threatening illness [6]. Although people without any clinical manifestations or with mild symptoms can be infectious, they usually go undetected by the healthcare systems and thus constitute a considerable challenge to disease surveillance and control [7,8]. Understanding variation within and between people in their contribution to onwards virus transmission is essential to improve the theory of DENV epidemiology, lead to innovation in outbreak detection and response, and refine assessments of prevention strategies [9].

The mosquito, Aedes aegypti, is the principal DENV vector and a primary driver of the increasing disease burden and expanding geographic distribution of dengue [1,10]. After an Ae. aegypti female takes a blood meal from a viremic human, virus transmission can happen if the virus successfully infects and replicates in the mosquito's midgut epithelium, crosses the midgut basal lamina, infects and replicates in the salivary glands, and is released in saliva as the mosquito bites another person. This process is generally represented by two distinct epidemiological parameters. Vector competence refers to the physiological ability of a mosquito to become infected and subsequently transmit the virus, whereas the extrinsic incubation period (EIP) represents the typical duration of this process [11]. It is well established that DENV vector competence is positively correlated with the virus dose ingested in the blood meal [7,12– 16]. Human infectiousness to mosquitoes roughly follows the kinetics of viremia, and symptomatic people can be infectious to mosquitoes from two days before to six days after the onset of symptoms [7,12-14,16,17]. Although human-to-mosquito DENV transmission is primarily determined by the magnitude of viremia, laboratory experiments indicate that vector competence and EIP also vary as a function of other factors such as the mosquito genotype, the virus strain, and the ambient temperature [18-23].

Several studies point to human factors other than viremia that influence DENV infectiousness to mosquitoes. A hospital-based study in Vietnam found that rising levels of DENV-specific antibodies were associated with a reduced likelihood of human-to-mosquito transmission independently of viremia [16]. A community-based study in Cambodia reported that asymptomatic and presymptomatic people were more infectious to mosquitoes than symptomatic people at the same level of viremia [7]. In laboratory studies, low-density lipoproteins [24,25] and serum iron [26] were shown to inhibit DENV acquisition by *Ae. aegypti* whereas blood glucose [27] and DENV non-structural protein 1 (NS1) antigenemia [28] enhanced mosquito infection. The blood concentration of some of these metabolites are known to vary within and between DENV-infected people [29,30] and could contribute to variation in DENV infectiousness to mosquitoes.

Regardless of the underlying mechanisms, knowledge of DENV transmission from humans to mosquitoes is epidemiologically imprecise and potentially misleading because it does not account for such variation in infectiousness within and between different people. This knowl-edge gap is due in large part to the logistical challenge of quantifying human-to-mosquito DENV transmission in a natural setting [31]. Here, we conducted a field-based study to characterize the temporal dynamics of viremic human infectiousness to mosquitoes with an emphasis on the difficult-to-study people with mild disease. Even though people with mild disease make up the bulk of human DENV infections, they are often missed by public health surveillance systems because surveillance programs are designed to identify infections in people who are ill enough to seek medical treatment [5]. We analyzed a dataset newly generated from a study design that we previously developed to identify DENV-infected volunteers with mild illness and then carried out direct mosquito feedings on those individuals [9,15,32]. Although our study focus was dengue, we also enrolled Zika virus (ZIKV)-infected participants during the study period when ZIKV circulated in Iquitos.

Methods

Ethics statement

The study protocol was approved by the U.S. Naval Medical Research Unit No. 6 (NAMRU-6) Institutional Review Board (Protocol Number NAMRU6.2014.0028), which includes Peruvian representation and complies with U.S. Federal and Peruvian regulations governing the protection of human subjects. Institutional Review Board authorization agreements were established between the U.S. NAMRU-6 and all participating institutions. The protocol was reviewed and approved by the Loreto Regional Health Department, which oversees health research in Iquitos. Written informed consent was obtained from all adult participants and the parents or guardians of all the child participants between 5–17 years. Participants that were minors provided assent (written for 8- to 17-years old, verbal for <8 years old).

Study design and sample testing

Study area and period. Iquitos is an urban center of ~400,000 inhabitants located in the Amazon Basin of northeastern Peru that functions almost as a self-contained epidemiological system because access to it is limited to air and river travel. The study area was previously described in detail [9]. DENV is endemic in Iquitos with a strong seasonal pattern. In general, DENV transmission is low from May to July, increases in August and September, and usually peaks between November and January. Our study was conducted from June 2015 to April 2019. In May 2016, the first human ZIKV infections were reported in Iquitos. ZIKV transmission continued through April 2017 with almost no detection of DENV during that time period.

Study protocol. The overall study design and procedures are described in detail by Morrison et al. [9]. Briefly, DENV and ZIKV cases were identified from two ongoing community-based cohort studies (~20,000 people with 3–5 febrile illness surveillance visits per week) initiated in June 2015 and from passive clinic-based disease surveillance carried out daily in approximately 6 facilities [33]. People identified with active DENV and ZIKV infections (i.e., detectable viremia) were invited to participate. Participation consisted of (1) direct mosquito feeding, (2) collection of venous blood samples, (3) health monitoring, and (4) symptom and movement questionnaires [9]. Study procedures were repeated daily if the participant's blood from the previous day tested positive for DENV or ZIKV and the participant agreed to continue. Participants could decline individual procedures on individual days and remain in the study.

Participant testing. Blood samples were generally tested within 24 hours of collection by reverse transcription quantitative PCR (RT-qPCR) for DENV and ZIKV RNA (see sample testing section below). As soon as results were available, study subjects who tested positive for DENV or ZIKV and had indicated willingness to participate in the viremic participant protocol were contacted and enrolled to perform the mosquito feeding and sequential blood sample collection procedures. During the ZIKV transmission period, Institutional Review Board permission was obtained to initiate the ZIKV-positive case protocol prior to obtaining the initial RT-qPCR result to maximize the probability of detecting human-to-mosquito transmission, because ZIKV-positive participants tended to be afebrile and were identified when they developed a rash. Participants were informed of their infection status within 12–24 hours and virus-positive participants were provided to study participants with an explanation by study personnel.

Mosquitoes. Human infectiousness was experimentally assessed using laboratory-reared mosquitoes derived from the local, wild-type *Ae. aegypti* population. To minimize

uncontrolled differences between experiments and maintain a level of genetic diversity representative of the wild mosquito population, a genetically diverse laboratory strain (GDLS) was created as previously described [9,34,35]. Briefly, the GDLS was established between August and December 2015 by collecting ~1,000 immature Ae. aegypti at each of 10 locations across Iquitos. Wild female progenitors were screened by RT-qPCR for DENV, the only known Aedes-borne virus circulating in Iquitos at the time, before hatching their eggs. None of the field-caught progenitors tested positive for DENV RNA. The 10 strains were maintained separately and merged to generate the GDLS mosquitoes used in blood-feeding experiments. All experiments were conducted within the first 6 generations of the GDLS. Virus-free mosquitoes were maintained in a field insectary constructed in a local household with a relative humidity (RH) of 70-80% and a temperature of approximately 27-28°C. The insectary had windows that let in natural light and room lights were turned on from approximately 0800 to 1700 each day. The natural photoperiod in Iquitos averages 12 hour:12 hour light:dark and variation between minimum and maximum daylight is approximately 1 hour. Mosquitoes that fed on viremic participants were held in a secure, laboratory-based insectary under the same temperature, RH, and light conditions. The GDLS mosquitoes were maintained with weekly noninfectious blood meals (chicken blood available for sale from a local butcher) through an artificial membrane. Resulting eggs were hatched in a water/tea (Collins Cinnamon tea) infusion (1 liter) for 24 hours. Two-hundred larvae per liter of tap water were placed in white 26.5×16.5 cm plastic pans. Larvae were provided a combination of wheat powder mixed with commercial fish food daily until pupation. Pupae were transferred to 1-pint plastic containers and placed into 1-gallon cardboard cages, provided with a moist towel to maintain humidity, a sugar cube, and a water pledget on top of the cages. As the mosquitoes emerged, females were separated into 1-pint cardboard cages containing 25 female Ae. aegypti each and 5 males to allow for mating. The workflow was adjusted to ensure that sufficient mosquitoes to carry out serial feeding experiments on up to 5 participants per week (60 mosquitoes per feeding event \times average of 3 feeding events per participant \times 5 participants = 900 per week). Sugar was removed from half of the cages on alternate days to ensure availability of sugar-starved mosquitoes on any given day.

Blood feeding. Mosquitoes were blood fed directly on the viremic participants according to a previously established procedure [9,15,32]. Briefly, two containers of 30 (15 for children <14 years of age) *Ae. aegypti* females from the GDLS were placed on the arms or legs of the participant for a maximum of 10 minutes in their homes. The option to carry out the feeding experiment in the U.S. Naval Medical Research Unit No. 6 (NAMRU-6) laboratory was offered as an alternative. Containers were then transported in a secure container to the NAMRU-6 laboratory for processing. Containers were placed in a -20° C freezer for about 1 minute to immobilize mosquitoes. Unfed mosquitoes were removed, and blood-engorged mosquitoes were kept in 1-pint cardboard containers with mesh tops, provided a sugar cube and sucrose solution soaked pledget, and held at 27°C and 70–80% RH for 7 to 16 days in a secure NAMRU-6 insectary. Mosquitoes were collected early (7–8 days) or late (12–16 days) relative to an expected EIP of about 10 days at 27°C [36]. The duration of extrinsic incubation was assigned randomly to containers. For most feeding events, mosquitoes were divided so that some individuals were tested early (7–8 days) and others from the same feeding event were tested later (12–16 days).

Vector competence. On the day of harvest, saliva was collected from ~20 blood-fed mosquitoes according to a previously described protocol [16] (see Fig 3 in [9]). Briefly, legs and wings were removed, and the proboscis was inserted into the end of a pipette tip containing a 1:1 solution of fetal calf serum and 30% sucrose. After 30 minutes, the solution in the pipette tip was immediately inoculated intrathoracically into 4 to 5 naïve (i.e., not previously bloodfed) recipient *Ae. aegypti*. After 7 days of incubation, inoculated mosquitoes that were still alive (typically 3–5) were harvested to extract total RNA. Recipient mosquitoes that shared a saliva inoculum from the same mosquito were pooled for RNA extraction. Aliquots of RNA extracted from each of the ~20 inoculated mosquito pools were combined and tested for DENV or ZIKV RNA by RT-qPCR. If the pool of RNA extracts was negative, the participant was scored not infectious to mosquitoes for that feeding event. If the pool of RNA extracts was positive, RNA from each of the ~20 inoculated mosquito pools was individually tested by RT-qPCR to determine the proportion of blood-fed mosquitoes with infectious saliva. If multiple feedings were performed on a single participant, each feeding event was considered independent of each other and pools were tested as described above.

Sample testing. Viral RNA was detected and quantified by RT-qPCR. RNA was extracted from whole blood and/or serum samples using QIAamp Viral RNA Mini Kit (Qiagen) following the manufacturer's guidelines. The same extraction method was used for all mosquito samples except for inoculated mosquitoes, from which DENV RNA was extracted using a crude extraction method previously described [37]. Briefly, mosquito samples were placed in a vial containing a 4-mm glass bead and homogenized with a TissueLyser (Qiagen) after adding 250 µl of squash buffer (10 mM Tris pH 8.2, 1 mM EDTA, 50 mM NaCl) and 15 mg/ml of proteinase K (Qiagen). The homogenate was then heated at 56°C for 5 minutes and 98°C for 10 minutes. DENV RNA was detected and quantified using a serotype-specific TaqMan RTqPCR assay based on a previously published method [38]. ZIKV RNA was detected and quantified using aTaqMan RT-qPCR assay targeting the envelope gene (forward primer: 5'- GAT CCTACTGCTATGAGGCATC-3'; reverse primer: 5'- CCAGCCTCTGTCCACTAACGT-3'; probe: 5'- FAM-AGCCTACCTTGACAAGCAGTCAGACACTCAA-BHQ-3') according to the following thermal profile: 50°C for 15 minutes, 95°C for 2 minutes, followed by 40 cycles of 95°C for 15 seconds and 58°C for 30 seconds. RT-qPCR assays were performed on the ABI 7500 real-time PCR platform (Applied Biosystems) using TaqMan Fast Virus 1-step RT-PCR master mix (Life Technologies) or AgPath-ID One-Step RT-PCR kit (Applied Biosystems) for DENV and Superscript III Platinum One-Step qRT-PCR kit (Invitrogen) for ZIKV. Human samples with borderline cycle threshold (Ct) values for DENV (Ct = 34-36) were confirmed positive or negative using previously extracted RNA and a nested RT-PCR protocol previously described [39]. For ZIKV, samples with borderline Ct values underwent repeat testing. The limits of detection for the DENV and ZIKV assays were 12 and 31 genome copies/µl, respectively.

Data analysis

General model considerations. Details of the model selection process are provided in S1 Text. All candidate regression models relating the probability of an individual blood-fed mosquito becoming infectious to covariates were generalized linear models or generalized linear mixed models, specifically binomial regressions with a logit link. All models were run in R 4.1.3 [40]. The potential functional forms of the relationships that were evaluated, depending on the covariate, were (a) linear, (b) binary, (c) non-parametric, (d) smooth non-linear, or (e) shape-constrained smooth non-linear. Working backwards from the most complex models, any regression that contained a shape-constrained smooth term (see Viremia section below for details) was evaluated using mosquito-level data (i.e., presence or absence of virus in saliva as determined by inoculation and subsequent testing of recipient mosquitoes) and fit using the shape-constrained additive model (scam) package [41]. In this, the response variable was thus 0 (virus-negative saliva) or 1 (virus-positive saliva). All other models were evaluated using summarized experiment-level data; i.e., the fraction of tested mosquitoes that had virus-

positive saliva for a particular participant-incubation combination. As such, the response variable was equivalent to k infectious mosquitoes out of n, where k and n were specific to a particular participant-incubation combination. Any model that did not contain a shape-constrained term, but did contain a smooth non-linear term (e.g., the potential for a non-linear relation-ship between day of illness and transmission probability) was fit using generalized additive models with the mgcv package [42]. Models that had no smooth terms, but did consider repeated observations from the same participant, were fit using generalized linear mixed models with a random effect by individual with the lme4 package [43].

Treatment of covariates. Investigations were initiated with the simplest model and the most complex model and increased or decreased complexity as a function of model fit, parsimony, and biological feasibility, with the goal to arrive at a single model that balanced these three components. For each covariate considered, the simplest and most complex models considered are detailed below, as well as, in one case, justification for constraining the functional form to adhere more closely to biological plausibility. When possible (i.e., when comparing nested models), model comparison was done using a likelihood-ratio test. Otherwise, when appropriate, models were compared using Akaike information criterion (AIC) [44]. It should be noted that due to different response variables, models fit with scam were not directly comparable to other models in terms of AIC. In many cases, however, the same model could be reformulated in scam to allow for direct comparison. In terms of parsimony, when two compared models had similar AIC values (within 2 units of each other [45]), the model without the shape constraint was selected over the model with one, and the linear model was selected over the non-linear one. Also in the interest of parsimony, all non-linear smooth terms were constructed with 3 interior knots and all shape-constrained smooth terms were constructed with 4 interior knots; the minimum allowed in the respective packages.

Viremia. Consistent with earlier similar studies [7,16], viremia was measured as viral RNA concentration in serum samples and \log_{10} transformed. The simplest relationship considered between viremia and the probability of a mosquito becoming infectious was linear. *A priori* there is an expectation that a higher viremia would result in a higher probability of transmission; this is an assumption supported by results reported in previous literature [7,16]. As such, for the most complex relationship considered, the potential for a shape-constrained relationship between viremia and the response variable was considered, specifically a non-decreasing, monotonic smooth relationship.

Day of illness. At the time of mosquito feeding, the number of days after symptom onset ranged from 1 to 5 with more than 10 observations for each of the first 5 days post symptom onset. The simplest relationship considered was linear, while the most complex was non-linear. No *a priori* assumption was made about the shape of the relationship. Other researchers found a general decline in the probability of transmission with day of illness [16], but because the current study participants were enrolled earlier in their viremic period, it was unclear if the patterns previously observed would translate directly to this setting. In terms of the non-linear functional form, two approaches were considered: (1) a smooth non-linear b-spline-based term and (2) a fully non-parametric functional form where each day of illness had its own effect fit. Ostensibly, the smooth term would be the more parsimonious approach while the non-parametric functional form could allow for direct comparison between different days' effects.

Extrinsic incubation. The duration of extrinsic incubation ranged from 7 to 16 days. *A priori* it was assumed that the probability of transmission would increase for longer incubation times. As such, the most complex functional form considered was a shape-constrained non-decreasing, monotonic smooth relationship. By design, experiments were either stopped early (7–8 days) or allowed to continue beyond EIP when most transmission is expected to occur

naturally (12–16 days). There was no sampling on days 9, 10, or 11 post blood feeding. In addition to the complex model described above, there were two simpler models considered, as well: (1) a linear relationship for incubation time or (2) a stepped relationship with one constant effect for 7–8 days of incubation and a different constant effect for 12–16 days of incubation. Both of these simpler models use a single degree of freedom, so in the absence of fit dominance, it would be difficult to identify the most parsimonious between them.

Repeated measures. Of the 27 participants included in the analysis, 9 contributed to only one experiment; i.e., a single feeding event resulting in a single incubation-day of illness combination. Fifteen contributed to two experiments, one contributed to three, and two contributed to four. While there is potential for the influence of repeated measures on the experiment outcome (e.g., a participant who is systematically more likely to infect mosquitoes than their viremia would indicate), there are relatively low numbers of total participants and repeatedly measured participants. Due to low sample sizes and concerns around convergence and interpretability, a random effect by participant was not included. Care must be taken to not overinterpret the conclusions of our analysis as the sample size is low.

Results

We identified people with active infections by detecting viral RNA in their blood, which we refer to as viremia hereafter. A total of 54 viremic people consented to blood feed mosquitoes directly 1-4 times during their infection, which resulted in 116 mosquito feeding events. All 54 study participants were symptomatic; 96.3% were febrile. Forty-three (79.6%) participants were infected with DENV-2, 2 (3.7%) with DENV-3, and 9 (16.7%) with ZIKV. Our analysis excluded events for which viremia was detectable at enrollment but undetectable at the time of mosquito feeding, which was the case for all events involving participants with DENV-3 and ZIKV infections. Five participants (2 DENV-3 and 3 DENV-2 infections) were infectious to mosquitoes despite having undetectable viral RNA in their blood at the time of feeding. In total, our analysis was conducted on 50 feeding events involving 27 participants (44.4% female) that had detectable DENV-2 viremia at the time of feeding. Of these feeding events, 58% were on participants <17 years of age, 30% on participants 18–31 years of age, and 12% on participants 40-80 years of age (S1 Fig). Corresponding primary data are provided in supplementary information (S1 File) and their descriptive statistics are shown in Table 1. Variation in the number of mosquitoes tested reflects variation in the number of mosquitoes available, variation in the blood feeding rate and/or variation in the mosquito survival rate during the incubation period.

We analyzed the probability of a mosquito becoming infectious (i.e., a direct measure of vector competence) using logistic regression models. The final selected model retained a linear effect of viremia, a non-linear smooth effect of day of illness, and a stepped effect of extrinsic

Table 1. Summary table of descriptive statistics. Summary statistics are provided for the main variables of the final dataset analyzed.

Variable	Range	Median	Mean	Std. dev.
Participant age (years)	5-80	17	22.4	17.0
Days of illness at the time of feeding	1-5	2	2.46	1.20
Number of feeding events per participant	1-4	2	1.85	0.82
Viremia at the time of feeding (log ₁₀ scale)	2.0-9.1	5.63	5.29	2.05
Number of mosquitoes tested	1-20	18.5	16.6	4.84
Percentage of infectious mosquitoes	0-45	0	8.76	13.6

https://doi.org/10.1371/journal.pntd.0011593.t001

Model term	Estimate	Std. error	z-value	<i>p</i> -value
Intercept	-7.0656	0.7934	-8.905	<2×10 ⁻¹⁶
Log viremia	0.3237	0.0831	3.895	0.000098
Incubation > 11 days	3.0541	0.6063	5.038	0.00000047
Model term	Eff. df	Ref. df	Chi ²	<i>p</i> -value
Day of illness	1.91	1.99	9.775	0.00565

Table 2. Parameters of the final selected model. The probability of a mosquito being infectious was analyzed using logistic regression models. The table shows the effects retained in the final selected model. Eff. df: effective degrees of freedom; Ref. df: reference degrees of freedom.

https://doi.org/10.1371/journal.pntd.0011593.t002

incubation time. While this model was not the best model in terms of AIC, no model had an AIC less than 2 units smaller and all models with smaller AIC were deemed biologically implausible (see S1 Text for details). Each term was statistically significant (Table 2). For each log₁₀-unit increase in viremia, the odds of a mosquito being infectious were 1.38 times higher (odds ratio [OR] = 1.38, *p*-value <0.0001, 95% confidence interval [CI]: 1.17–1.63). Extrinsic incubation for 12 days or greater increased the odds of a mosquito being infectious by 21.3 times relative to 7 or 8 days of incubation (OR = 21.3, *p*-value <0.00001, 95% CI: 6.46–69.58). Fig 1 shows ORs for each day of illness against the mean estimated odds for day of illness of 1.



Fig 1. DENV transmission probability is highest on the second day of illness. The odds of a mosquito being infectious for each day of illness are plotted on a \log_{10} scale relative to the mean estimated odds for the first day of illness. Vertical bars show the 95% confidence intervals of the odds ratios. The horizontal dotted line represents an odds ratio of 1.

https://doi.org/10.1371/journal.pntd.0011593.g001

Of particular note, the ORs for the second day of illness compared to the mean estimated effect for the first day of illness was 2.24 (95% CI: 1.56–3.21). This approach does not easily produce a *p*-value for this comparison, but the entire non-linear smooth model is statistically significant (*p*-value = 0.00565). Another model that resulted in a similar fit does allow for a direct calculation of the OR between the first and second days of illness. The model with no smooth terms, a linear effect of viremia, a non-parametric effect of day of illness, and a stepped effect of incubation time had an AIC of 139.52 ($\Delta AIC = 3.34$). That model, which is arguably simpler even though it uses more degrees of freedom, revealed extremely similar effects of viremia and incubation time. In that model, which fit the effect of each day of illness was 2.23 (*p*-value = 0.0062, 95% CI: 1.26–3.96). Fig 2 illustrates the fitted effects of the covariates on predicted probabilities across a wide range of potential values.

Discussion

In this study, we quantified human-to-mosquito DENV transmission from viremic people to determine the drivers of infectiousness to mosquitoes in a natural context. Our experimental setup allowed us to blood feed mosquitoes directly on human volunteers with mild illness during their DENV infection [9,15,32]. We measured the proportion of mosquitoes that became infectious (i.e., that had infectious virus in their saliva) and used regression analyses to identify the factors influencing this process. We confirmed that the magnitude of viremia is a significant predictor of the probability of a blood-fed mosquito becoming infectious. Consistent with earlier studies, a longer extrinsic incubation also increased transmission probability [46]. A result that is different from previous studies was that the day of illness influenced the probability of onward DENV transmission in a non-monotonic fashion with a maximum around 2 days after symptom onset. This finding challenges the previous belief that the maximum level of infectiousness to mosquitoes coincides with symptom onset [17].

Day of illness was previously found to negatively influence human-to-mosquito transmission of DENV-1 and DENV-2 from hospitalized patients enrolled 2–4 days after symptom onset [16]. For each unit increase in day of illness, the odds of a mosquito becoming infected were 0.6 times lower [16]. Our results reveal that the relationship between infectiousness and day of illness is more complex than a simple linear relationship. By using non-linear functional forms in our regression models, we found that the odds of DENV transmission on the second day of illness were 2.23 times higher than on the first day of illness. Infectiousness subsequently decreased until the fifth day of illness. One important implication of this finding is that virus transmission can be substantially reduced if mildly symptomatic DENV-infected people take protective measures against mosquito bites within the first two days of their illness. Protective measures could include personal protection (e.g., topical repellents) and/or insecticide spraying at the locations of daytime activity.

At this point, we can only speculate on the factors other than viremia that drive temporal variation in DENV infectiousness to mosquitoes. DENV-infected people experience most symptoms during their first week of illness, but there is substantial heterogeneity in the duration and reported intensity of individual symptoms [47]. Immune factors, metabolites or changes in blood concentration of viral antigens during infection could modulate mosquito infection [16,26,27,29,30]. The effect of human blood factors on vector competence could be due to their direct influence on mosquito-virus interactions or indirectly mediated by changes of the mosquito gut microbiota [48].

A limitation of our study is the small number of human participants that contributed to the analysis and the diminishing number of volunteers who participated in repeated feedings over



Fig 2. Human-to-mosquito DENV transmission depends on viremia, incubation time and day of illness. DENV transmission probability was estimated by the proportion of blood-fed mosquitoes that had infectious virus in their saliva. Each circle (12–16 days of incubation) or triangle (7–8 days of incubation) represents a single mosquito feeding event and the size of the circle or triangle is proportionate to the number of mosquitoes analyzed. Lines represent the regression fits to the data. Lines and data points are color and shape coded according to the inset scale. (a) Transmission probability as a function of \log_{10} -

transformed viremia. (b) Transmission probability as a function of day of illness. (c) Transmission probability as a function of extrinsic incubation time in days. (d) Log₁₀-transformed viremia as a function of day of illness.

https://doi.org/10.1371/journal.pntd.0011593.g002

time. We excluded 2 participants infected with DENV-3, all 9 participants infected with ZIKV, and 16 of the 43 participants with a DENV-2 infection from the final dataset because they did not have detectable viremia at the time of mosquito feeding. Due to the low number of participants retained in the analysis (n = 27), we could not account for a random effect of participant and our statistical power was limited. In addition, participants in our study did not represent the entire spectrum of disease manifestations. All of our study participants were mildly symptomatic, which is the case for the majority of DENV infections [5]. It would be helpful in the future to determine whether our conclusions extend to the entire disease spectrum, ranging from asymptomatic infections to people that experience severe dengue, and to DENV sero-types other than DENV-2.

An additional limitation of our study was to measure viremia as the concentration of viral RNA in blood, which is distinct from the concentration of infectious viral particles [49]. Variation in infectiousness to mosquitoes could reflect differences in the ratio of infectious particles over viral genome copies. Intriguingly, a few participants with undetectable viremia were infectious to mosquitoes, as was observed in our previous study [7]. This could reflect variation in viral titer between venous blood (used to measure viremia) and capillary blood (imbibed by mosquitoes) and/or between serum and whole blood.

It is notable that despite conducting feeding experiments on 9 participants infected with ZIKV, none were infectious to mosquitoes. Viremia was undetectable on the day of feeding for all ZIKV-infected participants, including four feeding experiments that took place as early as two days post symptom onset. This is consistent with the previous observation that ZIKV viremia post symptom onset can be too low to achieve a productive mosquito infection [50]. It is possible that effective human-to-mosquito ZIKV transmission primarily occurs prior to symptom onset. Additional studies are required to determine the kinetics of human-to-mosquito ZIKV transmission during the viremic period.

Our results shed new light on variation within and between different people in their contribution to onward DENV transmission. Our finding that mildly symptomatic people are most infectious to mosquitoes on the second day of their illness points to an overlooked aspect of the dynamics of DENV infectiousness to mosquitoes during the viremic period. The net contribution of an individual to DENV transmission is a function of infectiousness dynamics and the frequency of human-mosquito contacts. Exposure to mosquitoes is heterogeneous [51–53] and difficult to quantify due to the unknown effect of coupled heterogeneities [54]. Human-mosquito contacts also depend on symptom severity [55], which highlights the need to relate illness with infectiousness dynamics. Detailed longitudinal information at the individual level can be used to improve surveillance and outbreak response as well as help to better inform modeling assessments of DENV transmission dynamics and the relative public health impact of different intervention tools and strategies [9].

Supporting information

S1 Fig. Demographic characteristics of 50 DENV-2 viremic feedings evaluated for infectiousness to *Aedes aegypti* mosquitoes.

(TIF)

S1 File. Primary data set used for analysis. (CSV)

S1 Text. Model Selection Procedures. Contains Figs A-B and Tables A-C. **Fig A.** Flow chart of model selection process. **Fig B.** Linear model fitting for day of illness. The plot shows the residuals of the linear fit against day of illness. Thin lines represent the 25^{th} and 75^{th} percentiles and solid lines represent the median. **Table A.** GLM with linear terms for each covariate (AIC = 163.2932). **Table B.** GAM with nonlinear smooth term for day of illness and linear terms otherwise (AIC = 146.0056). Eff. df: effective degrees of freedom; Ref. df: reference degrees of freedom. **Table C.** GLM with day of illness as a factor and linear terms otherwise (AIC = 149.6279). (DOCX)

Acknowledgments

We thank the residents of Iquitos for their support and participation in this study. We are grateful to the Loreto Regional Health Department, including Drs. Hugo Rodriguez-Ferruci, Christian Carey, Carlos Alvarez, Hernan Silva and the Lic. Wilma Casanova Rojas, who all facilitated our work in Iquitos. W. Lorena Quiroz and Jhonny Cordova conducted dengue positive case follow-up and interviews and acknowledge the NAMRU-6 Iquitos febrile surveillance and laboratory teams in the rapid capture and diagnosis of dengue and Zika cases for direct mosquito feeding studies. We acknowledge the NAMRU-6 Virology and Emerging Infectious Disease and Command during the execution of this project. A special thanks to Gloria Talledo for her ongoing support with the preparation of protocols and reports for this project. We appreciate the commentary and advice provided by the NAMRU-6 Institutional Review Board and Research Administration Program for the duration of the study.

Author Contributions

- **Conceptualization:** Louis Lambrechts, Robert C. Reiner, Jr, Kanya C. Long, T. Alex Perkins, Alun L. Lloyd, Lance A. Waller, Uriel Kitron, Sarah A. Jenkins, Robert D. Hontz, Lauren B. Carrington, Cameron P. Simmons, John P. Elder, Valerie A. Paz-Soldan, Gonzalo M. Vazquez-Prokopec, Alan L. Rothman, Christopher M. Barker, Thomas W. Scott, Amy C. Morrison.
- **Data curation:** Robert C. Reiner, Jr, M. Veronica Briesemeister, Patricia Barrera, William H. Elson, Alfonso Vizcarra, Anna B. Kawiecki, Alan L. Rothman, Christopher M. Barker, Amy C. Morrison.
- Formal analysis: Louis Lambrechts, Robert C. Reiner, Jr, William H. Elson, T. Alex Perkins, Alun L. Lloyd, Gonzalo M. Vazquez-Prokopec, Alan L. Rothman.
- **Funding acquisition:** Louis Lambrechts, Robert C. Reiner, Jr, T. Alex Perkins, Alun L. Lloyd, Uriel Kitron, John P. Elder, Valerie A. Paz-Soldan, Gonzalo M. Vazquez-Prokopec, Alan L. Rothman, Thomas W. Scott, Amy C. Morrison.
- Investigation: Louis Lambrechts, M. Veronica Briesemeister, Patricia Barrera, William H. Elson, Alfonso Vizcarra, Helvio Astete, Isabel Bazan, Crystyan Siles, Stalin Vilcarromero, Mariana Leguia, Anna B. Kawiecki, Sarah A. Jenkins, Robert D. Hontz, J. Sonia Ampuero, Valerie A. Paz-Soldan, Gonzalo M. Vazquez-Prokopec, Alan L. Rothman, Thomas W. Scott, Amy C. Morrison.
- Methodology: Louis Lambrechts, Robert C. Reiner, Jr, M. Veronica Briesemeister, Patricia Barrera, Kanya C. Long, Mariana Leguia, T. Alex Perkins, Alun L. Lloyd, Uriel Kitron, Sarah A. Jenkins, Robert D. Hontz, Wesley R. Campbell, Lauren B. Carrington, Cameron P.

Simmons, John P. Elder, Gonzalo M. Vazquez-Prokopec, Alan L. Rothman, Christopher M. Barker, Thomas W. Scott, Amy C. Morrison.

- **Project administration:** Sarah A. Jenkins, Robert D. Hontz, Wesley R. Campbell, Gisella Vasquez, Valerie A. Paz-Soldan, Thomas W. Scott, Amy C. Morrison.
- **Resources:** Louis Lambrechts, Sarah A. Jenkins, Robert D. Hontz, Wesley R. Campbell, Thomas W. Scott, Amy C. Morrison.
- Software: Christopher M. Barker.
- Supervision: M. Veronica Briesemeister, Patricia Barrera, Helvio Astete, Isabel Bazan, Crystyan Siles, Stalin Vilcarromero, Mariana Leguia, Sarah A. Jenkins, Robert D. Hontz, J. Sonia Ampuero, Valerie A. Paz-Soldan, Thomas W. Scott, Amy C. Morrison.
- Visualization: Robert C. Reiner, Jr.
- Writing original draft: Louis Lambrechts, Robert C. Reiner, Jr, Kanya C. Long, Valerie A. Paz-Soldan, Thomas W. Scott, Amy C. Morrison.
- Writing review & editing: Louis Lambrechts, Robert C. Reiner, Jr, M. Veronica Briesemeister, Patricia Barrera, Kanya C. Long, William H. Elson, Alfonso Vizcarra, Helvio Astete, Isabel Bazan, Crystyan Siles, Stalin Vilcarromero, Mariana Leguia, Anna B. Kawiecki, T. Alex Perkins, Alun L. Lloyd, Lance A. Waller, Uriel Kitron, Sarah A. Jenkins, Robert D. Hontz, Wesley R. Campbell, Lauren B. Carrington, Cameron P. Simmons, J. Sonia Ampuero, Gisella Vasquez, John P. Elder, Valerie A. Paz-Soldan, Gonzalo M. Vazquez-Prokopec, Alan L. Rothman, Christopher M. Barker, Thomas W. Scott, Amy C. Morrison.

References

- 1. Brady OJ, Hay SI. The global expansion of dengue: How *Aedes aegypti* mosquitoes enabled the first pandemic arbovirus. Annu Rev Entomol. 2020; 65: 191–208.
- Kraemer MUG, Reiner RC Jr, Brady OJ, Messina JP, Gilbert M, Pigott DM, et al. Past and future spread of the arbovirus vectors Aedes aegypti and Aedes albopictus. Nat Microbiol. 2019; 4: 854–863. https:// doi.org/10.1038/s41564-019-0376-y PMID: 30833735
- Messina JP, Brady OJ, Golding N, Kraemer MUG, Wint GRW, Ray SE, et al. The current and future global distribution and population at risk of dengue. Nat Microbiol. 2019; 4: 1508–1515. https://doi.org/ 10.1038/s41564-019-0476-8 PMID: 31182801
- Messina JP, Brady OJ, Pigott DM, Golding N, Kraemer MUG, Scott TW, et al. The many projected futures of dengue. Nat Rev Microbiol. 2015; 13: 230–239. https://doi.org/10.1038/nrmicro3430 PMID: 25730702
- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, Moyes CL, et al. The global distribution and burden of dengue. Nature. 2013; 496: 504–507. https://doi.org/10.1038/nature12060 PMID: 23563266
- Simmons CP, Farrar JJ, van Vinh Chau N, Wills B. Dengue. N Engl J Med. 2012; 366: 1423–1432. https://doi.org/10.1056/NEJMra1110265 PMID: 22494122
- Duong V, Lambrechts L, Paul RE, Ly S, Lay RS, Long KC, et al. Asymptomatic humans transmit dengue virus to mosquitoes. Proc Natl Acad Sci U S A. 2015; 112: 14688–14693. <u>https://doi.org/10.1073/</u> pnas.1508114112 PMID: 26553981
- Ten Bosch QA, Clapham HE, Lambrechts L, Duong V, Buchy P, Althouse BM, et al. Contributions from the silent majority dominate dengue virus transmission. PLoS Pathog. 2018; 14: e1006965. <u>https://doi.org/10.1371/journal.ppat.1006965</u> PMID: 29723307
- Morrison AC, Paz-Soldan VA, Vazquez-Prokopec GM, Lambrechts L, Elson WH, Barrera P, et al. Quantifying heterogeneities in arbovirus transmission: Description of the rationale and methodology for a prospective longitudinal study of dengue and Zika virus transmission in Iquitos, Peru (2014–2019). PLoS One. 2023; 18: e0273798.
- Lambrechts L, Scott TW, Gubler DJ. Consequences of the expanding global distribution of Aedes albopictus for dengue virus transmission. PLoS Negl Trop Dis. 2010; 4: e646. <u>https://doi.org/10.1371/journal.pntd.0000646</u> PMID: 20520794

- Kramer LD, Ebel GD. Dynamics of Flavivirus infection in mosquitoes. Advances in Virus Research. Elsevier; 2003. pp. 187–232. https://doi.org/10.1016/s0065-3527(03)60006-0 PMID: 14689695
- Gubler DJ, Nalim S, Tan R, Saipan H, Sulianti Saroso J. Variation in susceptibility to oral infection with dengue viruses among geographic strains of Aedes aegypti. Am J Trop Med Hyg. 1979; 28: 1045– 1052. https://doi.org/10.4269/ajtmh.1979.28.1045 PMID: 507282
- Gubler DJ, Reed D, Rosen L, Hitchcock JR Jr. Epidemiologic, clinical, and virologic observations on dengue in the Kingdom of Tonga. Am J Trop Med Hyg. 1978; 27: 581–589. https://doi.org/10.4269/ ajtmh.1978.27.581 PMID: 677371
- Gubler DJ, Suharyono W, Lubis I, Eram S, Gunarso S. Epidemic dengue 3 in central Java, associated with low viremia in man. Am J Trop Med Hyg. 1981; 30: 1094–1099. <u>https://doi.org/10.4269/ajtmh. 1981.30.1094</u> PMID: 7283006
- Long KC, Sulca J, Bazan I, Astete H, Jaba HL, Siles C, et al. Feasibility of feeding Aedes aegypti mosquitoes on dengue virus-infected human volunteers for vector competence studies in Iquitos, Peru. PLoS Negl Trop Dis. 2019; 13: e0007116. <u>https://doi.org/10.1371/journal.pntd.0007116</u> PMID: 30753180
- Nguyet MN, Duong TH, Trung VT, Nguyen TH, Tran CN, Long VT, et al. Host and viral features of human dengue cases shape the population of infected and infectious Aedes aegypti mosquitoes. Proc Natl Acad Sci U S A. 2013; 110: 9072–9077. <u>https://doi.org/10.1073/pnas.1303395110</u> PMID: 23674683
- Nishiura H, Halstead SB. Natural history of dengue virus (DENV)–1 and DENV-4 infections: Reanalysis of classic studies. J Infect Dis. 2007; 195: 1007–1013. https://doi.org/10.1086/511825 PMID: 17330791
- Dickson LB, Merkling SH, Gautier M, Ghozlane A, Jiolle D, Paupy C, et al. Exome-wide association study reveals largely distinct gene sets underlying specific resistance to dengue virus types 1 and 3 in Aedes aegypti. PLoS Genet. 2020; 16: e1008794. https://doi.org/10.1371/journal.pgen.1008794 PMID: 32463828
- Fontaine A, Lequime S, Moltini-Conclois I, Jiolle D, Leparc-Goffart I, Reiner RC, et al. Epidemiological significance of dengue virus genetic variation in mosquito infection dynamics. PLoS Pathog. 2018; 14: e1007187. https://doi.org/10.1371/journal.ppat.1007187 PMID: 30005085
- Lambrechts L, Chevillon C, Albright RG, Thaisomboonsuk B, Richardson JH, Jarman RG, et al. Genetic specificity and potential for local adaptation between dengue viruses and mosquito vectors. BMC Evol Biol. 2009; 9: 160. https://doi.org/10.1186/1471-2148-9-160 PMID: 19589156
- Lambrechts L, Paaijmans KP, Fansiri T, Carrington LB, Kramer LD, Thomas MB, et al. Impact of daily temperature fluctuations on dengue virus transmission by *Aedes aegypti*. Proc Natl Acad Sci U S A. 2011; 108: 7460–7465.
- Pongsiri A, Ponlawat A, Thaisomboonsuk B, Jarman RG, Scott TW, Lambrechts L. Differential susceptibility of two field Aedes aegypti populations to a low infectious dose of dengue virus. PLoS One. 2014; 9: e92971. https://doi.org/10.1371/journal.pone.0092971 PMID: 24664142
- Watts DM, Burke DS, Harrison BA, Whitmire RE, Nisalak A. Effect of temperature on the vector efficiency of Aedes aegypti for dengue 2 virus. Am J Trop Med Hyg. 1987; 36: 143–152. <u>https://doi.org/10.4269/ajtmh.1987.36.143</u> PMID: 3812879
- 24. Tree MO, Londono-Renteria B, Troupin A, Clark KM, Colpitts TM, Conway MJ. Dengue virus reduces expression of low-density lipoprotein receptor-related protein 1 to facilitate replication in Aedes aegypti. Sci Rep. 2019;9. https://doi.org/10.1038/s41598-019-42803-9 PMID: 31015516
- Wagar ZL, Tree MO, Mpoy MC, Conway MJ. Low density lipopolyprotein inhibits flavivirus acquisition in Aedes aegypti. Insect Mol Biol. 2017; 26: 734–742.
- Zhu Y, Tong L, Nie K, Wiwatanaratanabutr I, Sun P, Li Q, et al. Host serum iron modulates dengue virus acquisition by mosquitoes. Nat Microbiol. 2019; 4: 2405–2415. <u>https://doi.org/10.1038/s41564-019-0555-x PMID</u>: 31527795
- Weng S-C, Tsao P-N, Shiao S-H. Blood glucose promotes dengue virus infection in the mosquito Aedes aegypti. Parasit Vectors. 2021;14. https://doi.org/10.1186/s13071-021-04877-1 PMID: 34311776
- Liu J, Liu Y, Nie K, Du S, Qiu J, Pang X, et al. Flavivirus NS1 protein in infected host sera enhances viral acquisition by mosquitoes. Nat Microbiol. 2016;1. https://doi.org/10.1038/nmicrobiol.2016.87 PMID: 27562253
- 29. Biswas HH, Gordon A, Nuñez A, Perez MA, Balmaseda A, Harris E. Lower low-density lipoprotein cholesterol levels are associated with severe dengue outcome. PLoS Negl Trop Dis. 2015; 9: e0003904. https://doi.org/10.1371/journal.pntd.0003904 PMID: 26334914

- Glasner DR, Puerta-Guardo H, Beatty PR, Harris E. The good, the bad, and the shocking: The multiple roles of dengue virus nonstructural protein 1 in protection and pathogenesis. Annu Rev Virol. 2018; 5: 227–253. https://doi.org/10.1146/annurev-virology-101416-041848 PMID: 30044715
- **31.** Carrington LB, Simmons CP. Human to mosquito transmission of dengue viruses. Front Immunol. 2014; 5: 290. https://doi.org/10.3389/fimmu.2014.00290 PMID: 24987394
- Morrison AC, Schwarz J, Long KC, Cordova J, Rios JE, Quiroz WL, et al. Acceptability of Aedes aegypti blood feeding on dengue virus-infected human volunteers for vector competence studies in Iquitos, Peru. PLoS Negl Trop Dis. 2019; 13: e0007090. https://doi.org/10.1371/journal.pntd.0007090 PMID: 30742621
- Forshey BM, Guevara C, Laguna-Torres VA, Cespedes M, Vargas J, Gianella A, et al. Arboviral etiologies of acute febrile illnesses in Western South America, 2000–2007. PLoS Negl Trop Dis. 2010; 4: e787. https://doi.org/10.1371/journal.pntd.0000787 PMID: 20706628
- Wise de Valdez MR, Nimmo D, Betz J, Gong H-F, James AA, Alphey L, et al. Genetic elimination of dengue vector mosquitoes. Proc Natl Acad Sci U S A. 2011; 108: 4772–4775. <u>https://doi.org/10.1073/pnas.</u> 1019295108 PMID: 21383140
- Wise de Valdez MR, Suchman EL, Carlson JO, Black WC. A large scale laboratory cage trial of Aedes densonucleosis virus (AeDNV). J Med Entomol. 2010; 47: 392–399. https://doi.org/10.1603/me09157 PMID: 20496587
- Tjaden NB, Thomas SM, Fischer D, Beierkuhnlein C. Extrinsic incubation period of dengue: Knowledge, backlog, and applications of temperature dependence. PLoS Negl Trop Dis. 2013; 7: e2207. https://doi. org/10.1371/journal.pntd.0002207 PMID: 23826399
- Yeap H, Axford JK, Popovici J, Endersby NM, Iturbe-Ormaetxe I, Ritchie SA, et al. Assessing quality of life-shortening Wolbachia-infected Aedes aegypti mosquitoes in the field based on capture rates and morphometric assessments. Parasit Vectors. 2014; 7: 58. https://doi.org/10.1186/1756-3305-7-58 PMID: 24495395
- Johnson BW, Russell BJ, Lanciotti RS. Serotype-specific detection of dengue viruses in a fourplex realtime reverse transcriptase PCR assay. J Clin Microbiol. 2005; 43: 4977–4983. <u>https://doi.org/10.1128/ JCM.43.10.4977-4983.2005 PMID: 16207951</u>
- Lanciotti RS, Calisher CH, Gubler DJ, Chang GJ, Vorndam AV. Rapid detection and typing of dengue viruses from clinical samples by using reverse transcriptase-polymerase chain reaction. J Clin Microbiol. 1992; 30: 545–551. https://doi.org/10.1128/jcm.30.3.545-551.1992 PMID: 1372617
- 40. R: A language and environment for statistical computing. R Foundation for Statistical Computing.
- 41. Pya N, Maintainer N. Package "scam." 2023. https://doi.org/10.1007/s11222-013-9448-7
- 42. Wood SN. Fast stable restricted maximum likelihood and marginal likelihood estimation of semiparametric generalized linear models. J R Stat Soc Series B Stat Methodol. 2011; 73: 3–36.
- 43. Bates D, Mächler M, Bolker B, Walker S. Fitting linear mixed-effects models Usinglme4. J Stat Softw. 2015;67. https://doi.org/10.18637/jss.v067.i01
- Akaike H. A new look at the statistical model identification. IEEE Trans Automat Contr. 1974; 19: 716– 723.
- 45. Burnham KP, Anderson DR. Multimodel inference. Social Methods Res. 2004; 33: 261–304.
- 46. Carrington LB, Tran BCN, Le NTH, Luong TTH, Nguyen TT, Nguyen PT, et al. Field- and clinically derived estimates of *Wolbachia* -mediated blocking of dengue virus transmission potential in *Aedes aegypti* mosquitoes. Proc Natl Acad Sci U S A. 2018; 115: 361–366.
- Elson WH, Reiner RC, Siles C, Bazan I, Vilcarromero S, Riley-Powell AR, et al. Heterogeneity of dengue illness in community-based prospective study, Iquitos, Peru. Emerg Infect Dis. 2020; 26: 2077– 2086. https://doi.org/10.3201/eid2609.191472 PMID: 32818402
- Caragata EP, Tikhe CV, Dimopoulos G. Curious entanglements: interactions between mosquitoes, their microbiota, and arboviruses. Curr Opin Virol. 2019; 37: 26–36. https://doi.org/10.1016/j.coviro. 2019.05.005 PMID: 31176069
- 49. Choy MM, Ellis BR, Gubler DJ, Ellis EM. Comparison of the mosquito inoculation technique and quantitative real time polymerase chain reaction to measure dengue virus concentration. Am J Trop Med Hyg. 2013; 89: 1001–1005. https://doi.org/10.4269/ajtmh.13-0100 PMID: 24019432
- 50. Tan C, Tan L, Hapuarachchi H, Lai Y, Wong P, Yap G, et al. Viral and antibody kinetics, and mosquito infectivity of an imported case of Zika fever due to Asian genotype (American strain) in Singapore. Viruses. 2018; 10: 44. https://doi.org/10.3390/v10010044 PMID: 29346272
- Perkins TA, Scott TW, Le Menach A, Smith DL. Heterogeneity, mixing, and the spatial scales of mosquito-borne pathogen transmission. PLoS Comput Biol. 2013; 9: e1003327. <u>https://doi.org/10.1371/journal.pcbi.1003327</u> PMID: 24348223

- Smith DL, Dushoff J, McKenzie FE. The risk of a mosquito-borne infectionin a heterogeneous environment. PLoS Biol. 2004; 2: e368. https://doi.org/10.1371/journal.pbio.0020368 PMID: 15510228
- Woolhouse ME, Dye C, Etard JF, Smith T, Charlwood JD, Garnett GP, et al. Heterogeneities in the transmission of infectious agents: implications for the design of control programs. Proc Natl Acad Sci U S A. 1997; 94: 338–342. https://doi.org/10.1073/pnas.94.1.338 PMID: 8990210
- Vazquez-Prokopec GM, Perkins TA, Waller LA, Lloyd AL, Reiner RC Jr, Scott TW, et al. Coupled Heterogeneities and Their Impact on Parasite Transmission and Control. Trends Parasitol. 2016; 32: 356–367. https://doi.org/10.1016/j.pt.2016.01.001 PMID: 26850821
- 55. Schaber KL, Perkins TA, Lloyd AL, Waller LA, Kitron U, Paz-Soldan VA, et al. Disease-driven reduction in human mobility influences human-mosquito contacts and dengue transmission dynamics. PLoS Comput Biol. 2021; 17: e1008627. https://doi.org/10.1371/journal.pcbi.1008627 PMID: 33465065